Author's response to reviews

Title: Assessment and model guided cancer screening promotion by village doctors in China: a randomized controlled trial protocol

Authors:

Rui Feng (853571123@qq.com)
Xingrong Shen (1264854421@qq.com)
Jing Chai (jingchai82@sina.com)
Penglai Chen (orangechenpl@aliyun.com)
Jing Cheng (jing_hp@sina.com)
Liang Han (1010066583@qq.com)
Ting Zhao (1252509750@qq.com)
Rui Sha (1214984721@qq.com)
Kaichun Li (295240260@qq.com)
Debin Wang (dbwang@vip.sina.com)

Version: 9 Date: 16 August 2015

Author's response to reviews: see over
Assessment and model guided cancer screening promotion by village doctors in China: a randomized controlled trial
Rui Feng, Xingrong Shen, Jing Chai, Penglai Chen, Jing Cheng, Liang Han, Ting Zhao, Rui Sha, Kaichun Li and Debin Wang

Dear editor,
Thank you very much for accepting the abovementioned manuscript for publication in BMC Cancer. We have revised the manuscript as suggested and enclosed please find our item by item responses to all issues raised.
With best regards,

Sincerely,

Debin Wang
School of Health Service Management
Anhui Medical University
China.

**Editorial suggestions and author responses**

Re: Please provide email addresses of all authors on the title page.
Author responses: We have provided the email addresses of all authors on the title page.

Re: Please provide Abbreviations section.
Author responses: We have added an abbreviations section (see lines 34 to 39).

Re: Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
Author responses: We added a statement that all authors have read and approved the manuscript at the end of the author contribution section (lines 376 to 377).